Notes of the RICH1 Phone meeting 27/10/2006
People Present:
CERN:  Fabio, Kenneth, Neville
IC:         Bill, Dave, Gabriel, Geoff(part)
Oxford:  Neville, Rohan
RAL:      Mike, Peter
Bristol:   Nick, Richard
Apologies for absence:
IC: Richard P.
Discussed Agenda Topics:
1. Gas Enclosure
The survey of the position of the HPDs windows has done on October 23rd 2006 (EDMS 788192). The damaged 0-ring was replaced. The GE movement restraints were put back on. The GE was opened up to remove the beampipe protection for the EW installation. The next steps are to clean the GE and do the leak testing.
2. Exit Window
Last Wednesday the safety inspection of the EW installation setup took place. The EW was installed last Thursday. The installation went smoothly. The EW was aligned centrally to the GE and not to the beampipe which needs to be aligned. The diaphragm was sealed onto the disk of the beampipe, but not onto the EW. Once the beampipe is re-aligned the EW will be aligned and the diaphragm fixed to it.

3. Beampipe
Need to get an estimate for the alignment timescale. The commisioning of the LHCb beampipe is in Feb06. There will be no access to the interior of RICH1 during Feb06. The mirrors will be installed in March06. 
4. Quartz Windows
The coating is planned for Dec06. 

Tooling: The RICH2 QW support frame drawings will be modified for RICH1 and will be manufactured by the same company as RICH2.

Storage: The RICH2 boxes are not big enough. Some other plastic boxes have been found.

5. Spherical Mirrors
Testing: The radiation testing is completed (10kGy ~10 years in LHCb). No change measured in the optomechanical properties of the CMA samples.  The other samples (Spherical Mirror Sample 1, Flat sample A and the demo mirror) are in the C4F10 gas tank and will be measured every ~3 months i.e. next measurement in Jan07. Also the optomechanical properties of these samples remained unchanged: demo mirror ~3 months C4F10, and small mirror + flat sample ~4 months C4F10.
Recoating: A CMA sample coated at CERN was uncoated and recoated. The measured reflectivity curve of the recoated sample remained unchanged proving that recoating of CMA substrates is feasible.  

Mechanics: Geoff has received form CMA the latest drawings of the frame. Geoff has to include in the drawings the survey marks and the planes with the fixing holes for the brackets. CMA will design the brackets.

CMA news: At last CMA phone meeting (18/10/06), CMA reported that the mandrel polishing is completed and R is within 0.1% of the nominal value. By the end of Nov06 the four mirrors will be ready for coating. On the 19/10/06 a CMA-Andre phone meeting took place to discuss the details of the coating procedure. After the meeting Andre sent CMA some information on how to improve the coating. We should receive 4 glass coated samples next week. The reflectivity, layer thicknesses and contaminants will be measured at CERN.  It has been decided to have regular CMA phone meetings every two weeks.
Coating at CERN: Calatroni’s lab has two chambers. 
1)1m diagonal chamber, but with an effective coating space of ~93cm diagonal. This chamber could be used for coating mirrors but the CMA mirrors do not fit (~83cm*63cm). 
 2) 1.5m diagonal (effective ~1.45m diagonal) used for sputtering; here CMA mirrors fit. 
To coat the CMA mirrors the equipment of the 1m chamber must be moved to the 1.5m chamber and vice versa. This involves ~1 month work.  The equipment is old and some parts might break and need replacement to work properly on the big chamber. Calatroni's group is busy with accelerator projects and exchanging the use of the two chambers (coating and sputtering) needs time and manpower which is not available. He advised to investigate coating the CMA mirrors at SESO in France and at Thales Optronics in Taunton-UK.
Coating outside CERN: The two companies have been contacted. Still waiting for a reply form Thales Optronics. SESO has la large coating chamber (~1.1m) and can make good coatings down to 160nm. A quote should arrive sometime next week.
If the new CMA samples reflectivity is not up to specs a decision on the coating has to be taken around the end of Nov06. 
6. Flat Mirrors
The drawings of the installation boxes should be ready by the end of Nov06. The manufacture of the installation boxes will be done through the CERN SOUS-TRAITANCE.  The alignment rig drawing requires more design work to include the top/bottom shelf for the spherical mirrors. Next Thursday at IC there will be a dedicated meeting to discuss these things.  
7. HPD Assembly
Two columns have been anodized and will be sent to CERN for the HPD dressing. The cooling manifold drawing is in an advanced state. The honeycomb has been ordered. The cable trays have been discussed with Rolf: all cables routed at x=0 (top/bottom). Work on the HPD magnetic shield insulation began two weeks ago (Vera, Maria, Richard P., and Tito). 30 shields have been fully tested. 180 shields have been wrapped; 18 more are required to complete RICH1, then a further 28 required for the two spare columns. Wrapping is nearly completed ~180. Testing will be carried out by Barbara at CERN.  The rail system concept is close to final. Next Monday there will be a meeting at IC to finalize the rail system. By the end of Nov06 the drawings will be ready.
8.  Magnetic Distortions

No news. Ray will be in Geneva beginning of Nov06 and a meeting will be arranged (action Dave).
9. Next meeting
Friday 10th November 2006 at 10:00(GMT) 11:00(CET). 
